Author's response to reviews

Title: Polygonum cuspidatum inhibits pancreatic lipase activity and adipogenesis via attenuation of lipid accumulation

Authors:
Young Sook Kim (ykim@kiom.re.kr)
Yun Mi Lee (candykong@kiom.re.kr)
Joo Hwan Kim (kimjh2009@gachon.ac.kr)
Jin Sook Kim (jskim@kiom.re.kr)

Version: 4 Date: 8 October 2013

Author's response to reviews:

Reviewer: Isao Matsui-Yuasa

Major comments
It is very important to show the data on the effect of each fraction on adipogenesis. The authors should add the data.

Response: As reviewer’s comments, we added the result on adipogenesis of each fraction in 3T3-L1 adipocytes (Fig. S1: page. 9 line 12~14 and page 20 line 1~10).